Author’s response to reviews

Title: Generic health literacy measurement instruments for children and adolescents: A systematic review of the literature

Authors:

Orkan Okan (orkan.okan@uni-bielefeld.de)
Ester Lopes (lopes.ester@outlook.com)
Torsten Bollweg (torsten.bollweg@uni-bielefeld.de)
Janine Bröder (janine.broeder@uni-bielefeld.de)
Melanie Messer (melanie.messer@uni-bielefeld.de)
Dirk Bruland (dirk.bruland@uni-bielefeld.de)
Emma Bond (e.bond@uos.ac.uk)
Graça Carvalho (graca@ie.uminho.pt)
Kristine Sørensen (contact@globalhealthliteracyacademy.org)
Luis Saboga-Nunes (saboga.nunes@gmail.com)
Diane Levin-Zamir (dianele@clalit.org.il)
Diana Sahrai (diana.sahrai@fhnw.ch)
Uwe Bittlingmayer (uwe.bittlingmayer@ph-freiburg.de)
Jürgen M. Pelikan (juergen.pelikan@univie.ac.at)
Malcolm Thomas (mlt@aber.ac.uk)
Ullrich Bauer (ullrich.bauer@uni-bielefeld.de)
Paulo Pinheiro (paulo.pinheiro@uni-bielefeld.de)

Version: 2 Date: 31 Dec 2017

Author’s response to reviews:

Dear BMC Editor,
The authors would like to thank you for your detailed review of our systematic review article on health literacy measurement tools for children and adolescents.

We have included the ‘Declarations’ heading and all associated sub-headings to the manuscript.

We hope to have met your expectations and thank you in advance for considering our manuscript for publication at BMC Public Health.

Best wishes,

Orkan Okan, on behalf of all authors